\documentclass[draft]{amsart}  % {{{
\usepackage[breaklinks, final]{hyperref}

% misc
\setlength{\parindent}{0pt} % no paragraph indents
\setlength{\parskip}{5pt}   % double space
\pagestyle{empty}           % no page numbers

% document specific stuff

%============================================================================}}}

\begin{document}
\title{EECS/MATH 510: Introduction to the theory of computing \\ Fall 2017}
\maketitle
\thispagestyle{empty}

\textbf{Professor:} Matthew Moore

\textbf{Office:} Eaton 2005C

\textbf{Office Hours:} T 10 -- 11, R 12:30 -- 13:30, and by appointment

\textbf{E-Mail:} \href{mailto:matthew.moore@ku.edu}
  {\nolinkurl{matthew.moore@ku.edu}}

\textbf{Course Website:} \url{http://www.ittc.ku.edu/~moore/comp/}

\textbf{Lecture Hours and Venue:} TR 11 -- 12:15, Summerfield Hall 506

\textbf{Textbook:} \emph{Introduction to Automata Theory, Languages, and
Computation} by John Hopcroft, Rajeev Motwani, and Jeffrey Ullman, 3rd ed.

\textbf{Homework:} There will be weekly homework assignments due every
Tuesday in class. Homework will consist of book problems as well as
``additional problems'', both of which will be posted on the course website.
Your lowest homework score will not be included in your grade.

\textbf{Quizzes:} There will be short weekly quizzes at the start of class
every Thursday. The quizzes will be over the material from the previous two
class session -- basic definitions and simple problems. Your lowest quiz
score will not be included in your grade.

\textbf{Exams:} There will be a midterm exam and a final exam. The midterm
exam will consist of an in-class portion and a take-home portion. The
in-class portion will be given on \textbf{October 10}. The take-home portion
of this exam will be due a week later. The date and time of the final exam
will be announced at a later date.

\textbf{Course Grading:} Your course grade will be calculated based on the
  breakdown below.
  \begin{center} \vspace{-0.5em} \begin{tabular}{ll} 
    Homeworks            & $45$\% \\
    Quizzes              & $15$\% \\
    Exams (incl.\ final)  & $40$\% 
  \end{tabular} \end{center} \vspace{-0.5em}
  Letter grades (including +/- designation) will be assigned as usual.

\textbf{Honor Code:} All work submitted for credit must be the student's
  own. If you have worked with another student on a homework problem, you
  should indicate it in your solution. Such cooperation is permitted, but
  your work must be your own. Quizzes and exams should be completed without
  unauthorized aid.

\textbf{Accommodation Procedure:} The Academic Achievement and Access Center
(AAAC) coordinates academic accommodations and services for all eligible KU
students with disabilities. If you have a disability for which you wish to
request accommodations and have not contacted the AAAC, please do so as soon
as possible. Information about their services can be found at
\url{http://www.access.ku.edu}. Please contact me privately in regard to
your needs in this course.

\textbf{Religious Observances:} Any student in this course who plans to
observe a religious holiday which conflicts in any way with the course
schedule or requirements should contact me as soon as possible to discuss
alternative accommodations.
\end{document}
